Author's response to reviews

Title: Role of Baseline Echocardiography Prior to Initiation of Anthracycline-based Chemotherapy in Breast Cancer Patients

Authors:

Alain Mina (aam34@mail.aub.edu)
Hind Rafei (hindrafei@gmail.com)
Maya Khalil (mayakk9@gmail.com)
Yasmin Hassoun (yasminhassoun@gmail.com)
Zeina Naser (znasser6@gmail.com)
Arafat Tfayli (at35@aub.edu.lb)

Version: 2 Date: 24 April 2014

Author's response to reviews: see over
Description of Revisions

1. A title page with authors and emails was added
2. Line and page numbering was added
3. Abstract modification to include background, methods, results, and conclusions
4. Competing interests section added (none)
5. Authors’ contributions section added
6. Acknowledgments section added

Maya Khalil, MD
Postdoctoral Research Fellow
Department of Internal Medicine
Naef K. Basile Cancer Center
American University of Beirut Medical Center
Tel: +9611-350000, ext: 7902

April 24, 2014